Development of a simple score based on HBeAg and ALT for selecting patients for HBV treatment in Africa

Graphical abstract



Highlights

- To eliminate hepatitis B, it is essential to scale up antiviral treatment programs.
- Access to conventional tools to assess treatment eligibility is limited.
- A new diagnostic score was developed using a large dataset of African patients.
- Diagnostic accuracy of the score for selecting patients for HBV treatment was high.
- The score may facilitate scale-up of treatment programs in resource-poor countries.

Authors

Yusuke Shimakawa, Ramou Njie, Gibril Ndow, ..., Maimuna Mendy, Mark Thursz, Maud Lemoine

Correspondence

yusuke.shimakawa@gmail.com (Y. Shimakawa), m.lemoine@imperial.ac. uk (M. Lemoine)

Lay summary

Limited access to the diagnostic tools used to assess treatment eligibility (liver biopsy/Fibroscan/hepatitis B virus DNA) has been an obstacle to the scale up of hepatitis B treatment programs in lowand middle-income countries. Using the data from African patients with chronic HBV infection, we developed and validated a new simple diagnostic score for treatment eligibility, which only consists of hepatitis B virus e antigen and alanine aminotransferase level. The diagnostic accuracy of the score for selecting patients for HBV treatment was high and could be useful in African settings.

http://dx.doi.org/10.1016/j.jhep.2018.05.024



Development of a simple score based on HBeAg and ALT for selecting patients for HBV treatment in Africa

Yusuke Shimakawa^{1,*}, Ramou Njie², Gibril Ndow^{3,4}, Muriel Vray^{1,5}, Papa Saliou Mbaye⁶, Philippe Bonnard⁷, Roger Sombié⁸, Jean Nana⁹, Vincent Leroy⁹, Julie Bottero¹⁰, Patrick Ingiliz¹¹, Gerrit Post^{11,12}, Bakary Sanneh¹³, Ignatius Baldeh¹³, Penda Suso³, Amie Ceesay³, Adam Jeng³, Harr Freeya Njai³, Shevanthi Nayagam⁴, Umberto D'Alessandro³, Isabelle Chemin¹⁴, Maimuna Mendy¹⁵, Mark Thursz⁴, Maud Lemoine^{4,*}

 ¹Unité d'Épidémiologie des Maladies Émergentes, Institut Pasteur, Paris, France; ²The Gambia Hepatitis Intervention Study, IARC, c/o MRC Unit, Fajara, The Gambia; ³Medical Research Council (MRC) Unit, The Gambia at the London School of Hygiene and Tropical Medicine, The Gambia;
 ⁴Department of Surgery and Cancer, Liver Unit, Imperial College London, UK; ⁵Unité d'Épidémiologie des Maladies Infectieuses, Institut Pasteur de Dakar, Dakar, Senegal; ⁶Département d'Hépato-gastroentérologie, Hôpital Principal, Dakar, Senegal; ⁷Hôpital Tenon, Paris, France;
 ⁸Département d'Hépato-gastroentérologie, Centre Hospitalier Universitaire Yalgado Ouédraogo, Ouagadougou, Burkina Faso; ⁹Université Grenoble Alpes, Grenoble, France; ¹⁰Infectious Disease Department, St Antoine Hospital, AP-HP, Paris, France; ¹¹Center for Infectiology, Berlin, Germany; ¹²Charité – Universitätsmedizin Berlin, Germany; ¹³National Public Health Laboratory, Banjul, The Gambia; ¹⁴INSERM U1052, CNRS UMR5286, Centre de Recherche en Cancérologie, Université Claude Bernard, Lyon, France; ¹⁵International Agency for Research on Cancer (IARC), Lyon, France

See Editorial, pages 765–766

Background & Aims: To eliminate hepatitis B virus (HBV) infection, it is essential to scale up antiviral treatment through decentralized services. However, access to the conventional tools to assess treatment eligibility (liver biopsy/Fibroscan[®]/ HBV DNA) is limited and not affordable in resource-limited countries. We developed and validated a simple score to easily identify patients in need of HBV treatment in Africa.

Methods: As a reference, we used treatment eligibility determined by the European Association for the Study of the Liver based on alanine aminotransferase (ALT), liver histology and/ or Fibroscan and HBV DNA. We derived a score indicating treatment eligibility by a stepwise logistic regression using a cohort of chronic HBV infection in The Gambia (n = 804). We subsequently validated the score in an external cohort of HBV-infected Africans from Senegal, Burkina Faso, and Europe (n = 327).

Results: Out of several parameters, two remained in the final model, namely HBV e antigen (HBeAg) and ALT level, constituting a simple score (treatment eligibility in Africa for the hepatitis B virus: TREAT-B). The score demonstrated a high area under the receiver operating characteristic curve (0.85, 95% CI 0.79–0.91) in the validation set. The score of 2 and above (HBeAgpositive and ALT \geq 20 U/L or HBeAg-negative and ALT \geq 40 U/L) had a sensitivity and specificity for treatment eligibility of 85% and 77%, respectively. The sensitivity and specificity of the



Conclusions: A simple score based on HBeAg and ALT had a high diagnostic accuracy for the selection of patients for HBV treatment. This score could be useful in African settings.

Lay summary: Limited access to the diagnostic tools used to assess treatment eligibility (liver biopsy/Fibroscan/hepatitis B virus DNA) has been an obstacle to the scale up of hepatitis B treatment programs in low- and middle-income countries. Using the data from African patients with chronic HBV infection, we developed and validated a new simple diagnostic score for treatment eligibility, which only consists of hepatitis B virus e antigen and alanine aminotransferase level. The diagnostic accuracy of the score for selecting patients for HBV treatment was high and could be useful in African settings.

© 2018 European Association for the Study of the Liver. Published by Elsevier B.V. This is an open access article under the CC BY-NC-ND license (http://creativecommons.org/licenses/by-nc-nd/4.0/).

Introduction

Viral hepatitis is a major global health problem. In 2013, an estimated 1.45 million people died from viral hepatitis.¹ This is the seventh leading cause of death worldwide, ranked higher than any of the major infectious agents: human immunodeficiency virus (HIV), tuberculosis and malaria. Of these hepatitisrelated deaths, most of them occur in low-income and middle-income countries (LMICs), and about half are attributable to hepatitis B virus (HBV) infection, causing cirrhosis and hepatocellular carcinoma (HCC).¹

In 2016, the World Health Organization (WHO) developed an ambitious strategy to eliminate viral hepatitis as a public health threat by 2030, aiming to reduce the incidence of chronic HBV infection by 90%, and its mortality by 65%.² The WHO also set



Keywords: Hepatitis B; Diagnostic score; Patient care management; Validation studies; Sensitivity and specificity; Africa; Elimination.

Received 23 October 2017; received in revised form 2 March 2018; accepted 2 May 2018; available online 1 July 2018

^{*} Corresponding authors. Addresses: Unité d'Épidémiologie des Maladies Émergentes, Institut Pasteur, Paris, 25–28 rue du Dr Roux, 75015 Paris, France. Tel.: +33 (0)1 40 61 38 87; fax: +33 (0)1 45 68 88 76 (Y. Shimakawa), or Department of Surgery and Cancer, Liver Unit, Imperial College London, UK, Norfolk Place, London W2 1NY, UK. Tel.: +44 (0) 20 33 12 52 12; fax: +44 (0) 20 77 24 93 69 (M. Lemoine).

E-mail addresses: yusuke.shimakawa@gmail.com (Y. Shimakawa), m.lemoine@ imperial.ac.uk (M. Lemoine).

JOURNAL OF HEPATOLOGY

a global target for treatment coverage in people with chronic HBV infection eligible for antiviral therapy from 8% (2015) to 80% (2030).³

To scale up and decentralize screening, clinical assessment and treatment services in LMICs, it is essential to develop simple and validated diagnostic tests that are feasible and affordable in these contexts.⁴ HBV screening using inexpensive rapid tests that accurately detect hepatitis B surface antigen (HBsAg) has performed well in community and outreach settings,^{5,6} and the cost of antiviral treatment should no longer be the main obstacle (<US \$ 50 per year).^{3,7} Nevertheless, the clinical evaluation of treatment eligibility after confirming positive HBsAg remains complex and expensive. Because the current antiviral treatment is lifelong for most people, and not all chronic HBV infections result in liver-related deaths, the international guidelines require selection of patients who potentially benefit from the antiviral therapy by evaluating three factors: viral replication, liver fibrosis and inflammation.^{8–10} However, the recommended tools to assess these conditions (nucleic acid test to measure HBV DNA levels, liver biopsy and Fibroscan[®] to evaluate fibrosis stage) are rarely accessible and affordable in LMICs.^{11,12}

We therefore developed and validated treatment eligibility in Africa for the hepatitis B virus (TREAT-B), a simple score based on basic laboratory tests widely available in peripheral laboratories in LMICs, without relying on HBV DNA, liver histopathology or Fibroscan. Using a well characterized population-based cohort of treatment-naïve chronic HBV infection in The Gambia, we first developed a diagnostic score using the European Association for the Study of the Liver (EASL) treatment criteria, based on the conventional reference tests (HBV DNA, liver histology, or Fibroscan) as reference standard.⁸ Then, we assessed its diagnostic accuracy in a large cohort of treatment-naïve African adults chronically infected with HBV living in Senegal, Burkina Faso or European countries. We finally compared its performance with that of the HBV DNA-free WHO treatment criteria intended to be used in LMICs,¹³ and the risk estimation for HCC in chronic hepatitis B (REACH-B) score including HBV DNA as one of the variables.¹⁴

Patients and methods

Derivation dataset

From December 2011 to January 2014, the Prevention of Liver Fibrosis and Cancer in Africa (PROLIFICA), the first screen-andtreat program for HBV mono-infected people in sub-Saharan Africa, invited all Gambian adults identified as HBsAg carriers, by a rapid test (Determine, Alere, USA; or OnSite Combo Rapid Test, CTK Biotech, USA), for clinical evaluation through community-based and blood bank screening.^{6,7} In addition, all individuals known to be HBsAg-positive through historical community-based sero-surveys conducted in rural Gambia were invited.¹⁵ Most of them were asymptomatic and unaware of their infection. After informed consent, HBsAg-positive participants underwent a standardized clinical staging, including fasting transient elastography (Fibroscan 402, Echosens, France),¹⁶ abdominal ultrasonography, hematology (Medonic SE-12613, Boule Medical AB, Sweden), biochemistry (VITROS 350 analyser, Ortho, USA), hepatitis B e antigen (HBeAg) (ETI-EBK Plus, Diasorin, Italy), and HBV DNA using an in-house real-time PCR (limit of detection: 50 IU/ml).¹⁷ A subset of participants underwent liver biopsy and histopathological evaluation was performed by two independent pathologists as described in our previous paper.¹⁸ We performed all the examinations on the same day except the biopsy which was done within three months. We excluded from the analysis participants with: decompensated cirrhosis; HCC; prior or current antiviral treatment for HBV; co-infection with hepatitis C (HCV), D (HDV) or HIV; pregnancy; or missing clinical or virological data. We also excluded patients whose liver stiffness measurement using Fibroscan was unreliable, defined as a ratio of IQR divided by liver stiffness measurement is \geq 7.1 kPa.¹⁹

Candidate predictors selected for model derivation

Of demographic, clinical, hematological, biochemical, and virological variables commonly used to assess the severity of HBV-related liver disease, we selected a priori the following as candidate predictors potentially adapted and accessible in resource-limited settings: age, sex, HBeAg, aspartate aminotransferase (AST), alanine aminotransferase (ALT), alkaline phosphatase (ALP), gamma-glutamyl transpeptidase (GGT), albumin, total bilirubin and platelet count. $^{\tilde{8},15,20-22}$ Three biomarker-based fibrosis tests potentially useful in LMICs (AST-to-platelet ratio index [APRI]; fibrosis-4 score [FIB-4] and GGT-to-platelet ratio [GPR])^{13,18} were not included in the model to avoid the collinearity, because the components of these tests (age, AST, ALT, GGT and platelet count) were all independently assessed in the model. Alcohol consumption was not selected for a model, because no study subject had excessive alcohol intake >20 g/day in the Gambian cohort using a standardized questionnaire. Family history of liver cancer was also omitted as only a few individuals reported a positive history, probably related to poor diagnosis of HCC in the country.²³

International treatment guidelines

The antiviral treatment criteria given by each of the international guidelines are summarized (Table S1). Those of the EASL,⁸ American Association for the Study of Liver Diseases (AASLD),⁹ and Asian Pacific Association for the Study of the Liver (APASL)¹⁰ largely depend on viral load measurement and ALT level and/or fibrosis staging by liver histopathology or Fibroscan. For these criteria, we defined significant fibrosis and cirrhosis as Metavir \geq F2 and F4 in those who undertook liver biopsy, and liver stiffness \geq 7.9 kPa and \geq 9.5 kPa in those without biopsy based on the previous validation study,¹⁸ respectively. In contrast, the WHO guidelines provide HBV treatment criteria for LMICs without access to HBV DNA measurement: (i) cirrhosis diagnosed by physical examination or APRI >2.0 or (ii) persistently elevated ALT, without measuring HBV DNA.¹³ Because we primarily used cross-sectional data in this analysis, we considered the eligibility based on a single time point. In a subset of patients who had 2nd ALT measurement within six months before undergoing nucleos(t)ide analogue therapy, we performed a sensitivity analysis to assess the WHO treatment eligibility based on persistently elevated ALT at two consecutive visits. We applied the upper limits of normal for ALT as 30 IU/L for men and 19 IU/L for women, as recommended by the AASLD and WHO.^{9,13}

External validation set

To validate the new score, we analyzed historical datasets from hospital-based cross-sectional studies of treatment-naïve HBV mono-infected African adults consecutively recruited in six centers in five countries: Dakar, Senegal;²⁴ Ouagadougou, Burkina Faso;²⁵ Berlin, Germany; Paris and Grenoble, France; and Lon-

Research Article

don, UK. We only included the study participants who had complete data on hematology, biochemistry, HBeAg, HBV DNA and Fibroscan, performed on the same day, and applied the same exclusion criteria as the derivation set.

Statistical analyses

1

To identify predictors of the EASL treatment eligibility, univariable logistic regression was computed for each of the pre-selected candidate predictors. Continuous variables were transformed into logarithmic scale. A multiple logistic regression model was then fitted by including all the factors associated with the EASL criteria in the univariable analysis (two-sided p < 0.2). The final model was selected using the backward stepwise regression based on Wald test. The risk score directly derived from the multiple logistic regression model is complex and not practical for daily use. Therefore, we developed a simple point system without the need for calculator, by converting regression coefficients into integer points, and assigning these points to each level of each predictor.²⁶ From the total point, the probability to meet the EASL criteria was estimated with the equation:

$$1 + \exp(-\sum_{i=0}^{p} \beta_i W_i - B(Total \ point))$$
(1)

where β_i is the regression coefficient for the *i*th covariate; *Wi* is the reference value of the base category for the *i*th covariate; and *B* is the constant.²⁶

The performance of the newly developed diagnostic score was assessed in terms of calibration and discrimination in both derivation and validation sets for each of the international guidelines (EASL, AASLD and APASL) as a reference. The calibration was investigated by plotting the observed proportions of patients eligible for treatment against the predicted probabilities for each total point.²⁷ The correlation between the observed proportions and the predicted probabilities was assessed using Pearson's correlation coefficient.²¹ The capability of the score to correctly discriminate between those eligible and non-eligible for antiviral therapy was evaluated by using the receiver operating characteristic (ROC) curve. The optimal cut-off for the new score was selected to maximize the sum of sensitivity and specificity. The discrimination capabilities of the new score were compared to those of the WHO criteria¹³ and the REACH-B score¹⁴ using the area under the ROC curve (AUROC) for each of the reference international guidelines. The agreement between the new score and each of the international guidelines was also estimated using the prevalence-adjusted bias-adjusted kappa (PABAK).²⁸ To assess the need for multiple ALT measurements (as recommended by the WHO),¹³ we also evaluated the correlation and agreement of ALT levels measured between the first and second visits in a subset of Gambian patients who had a second ALT measurement, using Pearson's correlation coefficient and Bland-Altman plot, respectively. Finally, the performance of the new score was also assessed in a subgroup of patients defined by age, HBV genotype, HBeAg, presence of obesity and cirrhosis. All the analyses were performed using STATA 13.0 (Stata Corporation, USA). The study was approved by the Gambian Government/MRC Joint Ethics Committee, and reported in accordance with the STARD.²⁹

Results

Study participants

A total of 950 individuals with chronic HBV infection were enrolled in the PROLIFICA study in The Gambia. After exclud-

Viral Hepatitis



Fig. 1. Flow diagram of study participants. HBV, hepatitis B virus; HCV, hepatitis C virus; HDV, hepatitis D virus; HIV, human immunodeficiency virus; LSM, liver stiffness measurement.

ing those co-infected with HCV, HDV or HIV, with unreliable liver stiffness measurement and missing data, 804 were included in the derivation set (Fig. 1). Mean age was 38 years (SD ± 11) and 64% were men. For the validation set, 327 African people with chronic HBV infection were finally included in the analysis, and their study center was located as below: Senegal (n = 171), Burkina Faso (n = 36), Berlin (n = 34), Paris (n = 47), Grenoble (n = 23), and London (n = 16). They were younger (mean age 33 years, $SD \pm 10$) and with more men (76%) than the derivation set. Because of the difference in sampling methods (population-based in The Gambia vs. hospitalbased study in the validation set), the participants in the derivation set tended to have mild liver disease compared to those in the validation set (Table 1): the mean (±SD) HBV DNA level was $2.4 \pm 1.3 \log 10$ IU/ml in the derivation and 3.9 ± 1.5 log10 IU/ml in the validation set, and the prevalence of cirrhosis was 3% (24/804) and 12% (40/327), respectively. The proportion of participants eligible for antiviral therapy in the derivation set was similar across the different criteria (7% for EASL and AASLD, and 10% for APASL), apart from the WHO criteria for LMICs (49%). A similar difference in the proportion of eligible subjects was observed in the validation set between the WHO criteria without HBV DNA (65%) and the others (18–20%) (Table 1).

Development of the new diagnostic score: TREAT-B

In univariable analysis, the following candidate predictors were found to be significantly associated with the treatment eligibility according to the EASL guidelines: male sex, positive HBeAg, AST, ALT, ALP, GGT, and platelet count (Table 2). Subsequent multivariable analysis using backward stepwise procedures identified positive HBeAg and ALT level as independent predictors of the EASL treatment eligibility, with the following logistic regression model: risk score = $-13.0302 + (2.2052 \times HBeAg) +$ $(2.8755 \times \ln[ALT IU/L])$, where HBeAg was coded 0 for negative and 1 for positive. Based on the final regression model, we developed TREAT-B, a simple score by converting regression coefficients of HBeAg and ALT into integer points (Table 3).²⁶ The total point of TREAT-B was obtained by adding HBeAg score, negative (0 point) or positive (1 point), and ALT score, <20 IU/L (0 point), 20–39 (1 point), 40–79 (2 points) or ≥80 (3 points). TREAT-B ranged from 0 (HBeAg-negative and ALT <20 IU/L) to 4 (HBeAg-positive and ALT \geq 80 IU/L).

JOURNAL OF HEPATOLOGY

Table 1. Characteristics of study participants in the de	erivation set (n = 804) and validation set (n = 327).
--	---

		• •	
	Derivation set (n = 804)	Validation set (n = 327)	p value"
Age (years)	38 ± 11	33 ± 10	< 0.001
Age ≥40 years, n (%)	260 (32)	78 (24)	0.004
Male sex, n (%)	512 (64)	250 (76)	< 0.001
BMI (kg/m ²)	23 ± 5	23 ± 4	0.1
Obesity (BMI \geq 30 kg/m ²), n (%)	54 (7)	15 (6)	0.8
HBeAg, n (%)	45 (6)	35 (11)	0.002
HBV DNA (IU/ml)			
<2,000 (IU/ml)	704 (87)	98 (30)	< 0.001
2,000-20,000 (IU/ml)	38 (5)	150 (46)	
≥20,000 (IU/ml)	62 (8)	79 (24)	
Liver fibrosis			
No or mild (F0−1 or LSM ≤7.8 kPa)	730 (91)	280 (86)	< 0.001
Significant (F2–3 or LSM 7.9–9.4 kPa)	50 (6)	7 (2)	
Cirrhosis (F4 or LSM ≥9.5 kPa)	24 (3)	40 (12)	
AST (IU/L)	34 ± 30	39 ± 30	0.02
ALT (IU/L)	31 ± 30	43 ± 50	< 0.001
GGT (IU/L)	33 ± 33	44 ± 98	0.004
Albumin (g/L)	42 ± 4	42 ± 4	0.3
Total bilirubin (IU/L)	11 ± 7	11 ± 13	0.4
Platelets (10 ⁹ /L)	201 ± 68	207 ± 62	0.1
APRI	0.55 ± 0.75	0.48 ± 0.60	0.2
FIB-4	1.42 ± 1.72	1.12 ± 0.93	0.005
GPR	0.33 ± 0.45	0.57 ± 2.23	0.007
REACH-B score	4.3 ± 2.5	6.3 ± 2.7	<0.001
Eligible for EASL treatment criteria, n (%)	58 (7)	58 (18)	<0.001
Eligible for AASLD treatment criteria, n (%)	56 (7)	65 (20)	< 0.001
Eligible for APASL treatment criteria, n (%)	77 (10)	64 (20)	< 0.001
Eligible for WHO treatment criteria for LMICs, n (%)	393 (49)	214 (65)	< 0.001

AASLD, American Association for the Study of Liver Diseases; ALT, alanine aminotransferase; APASL, Asian Pacific Association for the Study of the Liver; APRI, AST-to-platelet ratio index; AST, aspartate aminotransferase; BMI, body mass index; EASL, European Association for the Study of the Liver; FIB-4, fibrosis 4; GGT, gamma glutamyl-transferase; GPR, GGT-to-platelet ratio score; HBeAg, hepatitis B e antigen; HBV, hepatitis B virus; LMICs, low- and middle-income countries; REACH-B, risk estimation for hepatocellular carcinoma in chronic hepatitis B; WHO, World Health Organization.

^{*}Continuous variables are presented as mean ± SD.

** p values were obtained using t-test for continuous variables, and chi-squared test for categorical variables.

¹ Liver fibrosis was staged by histopathology in 107 participants who undertook liver biopsy, and by Fibroscan in the rest of the participants.

Validation

The proportion of eligible patients according to each of the international guidelines, by the total point of TREAT-B are presented (Fig. 2 and Table S2). In both the derivation and validation sets, the proportion meeting the treatment criteria increased with increasing total point. The predicted probability of meeting the treatment criteria increased steadily: 0.4%, 3.0%, 19.6%, 65.2%, and 93.5% for a total point of 0, 1, 2, 3, and 4, respectively (Fig. 3). The calibration plot (Fig. 3 for EASL, and Fig. S1 for AASLD and APASL) showed that the points fell close to the 45° line (perfect calibration),²⁷ and the correlation coefficients were high: 0.998 for EASL, 0.994 for AASLD and 0.993 for APASL in the validation set.

TREAT-B showed a good discrimination capability for EASL treatment criteria with AUROC of 0.88 (95% CI 0.83–0.93) and 0.85 (0.79–0.91) in the derivation and validation set, respectively (Table 4, Fig. 4). In addition, TREAT-B also performed well for AASLD and APASL treatment guidelines: AUROC was 0.89 (0.84–0.94) and 0.83 (0.77–0.89) for AASLD, and 0.87 (0.83–0.91) and 0.85 (0.80–0.90) for APASL, in the derivation and validation set, respectively. The sum of sensitivity and specificity was highest with a cut-off of 2 points; the sensitivity and specificity to indicate EASL, AASLD, and APASL treatment eligibility criteria in the validation set were 85% and 77%, 82% and 78%, 83% and 78%, respectively (Table 4). The sensitivity and specificity by different cut-off points are presented (Table S3).

Comparison with the WHO criteria

The performance of the TREAT-B score was compared with the WHO criteria for LMICs, which rely on ALT and APRI without HBV DNA. To select African patients infected with HBV for antiviral therapy, the AUROCs of TREAT-B were significantly higher than those of the WHO criteria (Table 4, Fig. 4). The WHO criteria demonstrated high sensitivities (ranging 86–94%), but inadequately low specificities (40–56%). PABAK, the measure of agreement between the TREAT-B score and each of the international guidelines, was also high (ranging from 0.57 to 0.76), compared to that of the WHO criteria (ranging from -0.03 to 0.19) (Table 4).

In a subset of Gambian patients who had available data on 2nd ALT measurement (n = 472), a sensitivity analysis was performed by using persistently elevated ALT levels at two consecutive measurements as one of the WHO criteria. While only 10.4% (49/472) were eligible according to the EASL criteria, a total of 162 patients (34.2%) were categorized as eligible using the WHO treatment criteria, resulting in a false positive rate of 29.6%. Significantly higher AUROCs of TREAT-B than the WHO criteria were also confirmed in this sensitivity analysis (Table S4). The correlation and agreement of ALT levels measured between the 1st and 2nd ALT measurement were good in this subgroup (Fig. S2).

Comparison with the REACH-B

To investigate whether having HBV DNA significantly improves the performance of the score, we compared the AUROCs of Table 2. Predictors for the treatment eligibility in the derivation set from The Gambia (n = 804).

Variables	Not eligible for	Eligible for treatment	Crude <i>p</i> value ^{**}	Final model selected by backward stepwise regression ""			
	treatment (n = 746) [*]	by EASL guidelines (n = 58)*		Regression coefficient	p value		
Age (years)	38 ± 11	36 ± 13	0.2				
Male sex, n (%)	466 (62)	46 (79)	0.01				
HBeAg, n (%)	22 (3)	23 (40)	< 0.001	2.2052	< 0.001		
AST (ln IU/L)	3.4 ± 0.3	4.0 ± 0.6	< 0.001				
ALT (ln IU/L)	3.2 ± 0.4	4.1 ± 0.8	< 0.001	2.8755	< 0.001		
ALP (In IU/L)	4.5 ± 0.3	4.6 ± 0.4	< 0.001				
GGT (ln IU/L)	3.3 ± 0.5	3.9 ± 0.8	< 0.001				
Albumin (ln g/L)	3.7 ± 0.1	3.7 ± 0.1	0.7				
Total bilirubin (ln IU/L)	2.3 ± 0.5	2.4 ± 0.4	0.7				
Platelets (ln 10 ⁹ /L)	5.3 ± 0.4	5.1 ± 0.4	< 0.001				

ALP, alkaline phosphatase; ALT, alanine aminotransferase; AST, aspartate aminotransferase; EASL, European Association for the Study of the Liver; GGT, gamma glutamyltransferase; HBeAg, hepatitis B e antigen.

^{*}Continuous variables were log-transformed (except age) and were presented as mean ± SD.

^{**}Crude *p* values were obtained using *t* test for continuous variables after logarithmic transformation, and chi-squared test for categorical variables.

**** Final regression model = -13.0302 + (2.2052 * HBeAg) + (2.8755 * ln(ALT)).

TREAT-B with those of REACH-B, composed of sex, age, ALT, HBeAg and HBV DNA.¹⁴ Both demonstrated similar AUROCs without any significant difference (Table 4, Fig. 4).

Performance of TREAT-B in subgroup of patients

As presented (Table S5), the performance of TREAT-B did not vary according to the age groups, the different HBV genotypes, the presence of HBeAg, obesity (BMI $\ge 30 \text{ kg/m}^2$), or cirrhosis. The AUROCs were: 0.86 (95% CI 0.81–0.91) for <40 years and 0.92 (0.87–0.97) for ≥ 40 years; 0.88 (0.63–1.00) for genotype A and 0.87 (0.81–0.94) for genotype E; 0.83 (0.77–0.88) for HBeAg-negative and 0.83 (0.75–0.90) for HBeAg-positive; 0.86 (0.82–0.91) for BMI <30 kg/m² and 0.98 (0.96–1.00) for $\ge 30 \text{ kg/m}^2$; and 0.88 (0.82–0.93) for those without cirrhosis and 0.86 (0.71–1.00) with cirrhosis.

Discussion

This is the first study that developed and validated a diagnostic prediction score for treatment eligibility in African individuals with chronic HBV infection. Using a unique population-based cohort of Gambian HBV-infected individuals, we developed a score free from HBV DNA (TREAT-B) on the basis of ALT level and HBeAg sero-status. In a large external cohort of African patients living in West Africa or Europe, we confirmed the high diagnostic accuracy of the new score. TREAT-B correctly identified 82-85% of patients meeting the international treatment criteria based on the conventional reference tests (serum HBV DNA and liver biopsy or Fibroscan), and 77-78% of patients who do not fulfil these reference criteria. Moreover, TREAT-B performed well irrespective of the age groups, HBV genotypes, the presence of obesity, or cirrhosis, and better than the WHO treatment criteria. The score including HBV DNA (i.e., REACH-B) did not perform better than the TREAT-B, supporting the usefulness of TREAT-B even in a context where HBV DNA measurement is accessible.

In HBV-infected people the level of serum HBV DNA has been one of the most important predictors for the development of HBV-related liver diseases, including HCC,^{20–22} and has been used as a key marker for deciding the eligibility for antiviral therapy.^{8–10} However, real-time PCR, the current standard assay to measure viral load, is not accessible and affordable for the majority of people living in LMICs; the assay is costly (US\$ 60-200 per test), requires sophisticated facilities, equipment and technicians with specialized training, and thus its availability is often limited to reference laboratories in large urban centres.^{3,4} In Madagascar, for example, there are numerous laboratories equipped with immunoassays for HBV serology, however, no facility performs HBV DNA PCR, and patients' sera need to be shipped to a commercial laboratory in Europe when HBV DNA measurement is requested by a hepatologist.¹² Consequently, on the basis of expert opinion, the WHO recently recommended, in settings where HBV DNA assay is unavailable, to treat individuals with persistently abnormal ALT levels (defined by ALT over the upper limit of normal at three different time points) irrespective of HBeAg status.¹³ However this recommendation is problematic since it requires several blood tests and medical visits before the treatment decision can be made, which may eventually impede the patients' retention in care. In our analysis, we found moderate correlation and agreement of ALT levels measured at two consecutive visits within a six-month period, which may support the clinical evaluation at a single point in time for the treatment eligibility. Moreover, by applying persistently elevated ALT levels at two consecutive measurements, we found a low specificity of the WHO criteria (70%), which would lead to unnecessary lifelong treatment in 30% of patients who do not meet the reference treatment criteria. Similar discrepancy between the treatment eligibility criteria based on the WHO guidelines and those based on other international guidelines has been reported from other African countries.^{30,31}

Natural history of chronic HBV infection differs remarkably between Africa and Asia. In contrast to East Asia where half of children with chronic HBV infection remain positive for HBeAg into their adulthood, spontaneous loss of HBeAg occurs much faster in Africa, where only 10% remain HBeAg-positive in their twenties.^{15,32} This geographic variation is often explained by the difference in viral genotypes, but also by the difference in the major mode of HBV transmission (mother-to-child transmission in Asia and early horizontal transmission in Africa).^{15,33} Nevertheless, in both regions HBeAg has been constantly found to be an important predictor of HBV-related liver disease.^{34,35} In our derivation and validation cohort, 51.1% and 54.3% of HBeAgpositive participants were eligible for the EASL treatment criteria, compared to 4.6% and 13.4% of HBeAg-negative participants, respectively.

JOURNAL OF HEPATOLOGY

Predictors	Categories	Reference value [*]	Regression coefficient	Regression units	Points
HBeAg	Negative	0	2.2052	0 (base category)	0
	Positive	1		2.2052	1
ALT (IU/L)	<20	2.62	2.8755	0 (base category)	0
	20-39	3.33		2.0416	1
	40-79	4.02		4.0257	2
	≥80	4.73		6.0673	3

Table 3. Development of TREAT-B points system based on the selected model (HBeAg and ALT).

ALT, alanine aminotransferase; HBeAg, hepatitis B e antigen; TREAT-B, treatment eligibility in Africa for the hepatitis B virus.

^{*} Reference value is presented in natural logarithmic scale for ALT.



Fig. 2. Proportion of patients eligible for treatment according to each of the international guidelines, by the total point (0–4) of TREAT-B. The number of patients with the TREAT-B score of 0, 1, 2, 3, and 4 was 214, 457, 93, 33, and 7 in the derivation set, and 32, 185, 68, 36, and 6 in the validation set, respectively. AASLD, American Association for the Study of Liver Diseases; APASL, Asian Pacific Association for the Study of the Liver; EASL, European Association for the Study of the Study of the Association for the Study of the Liver; TREAT-B, treatment eligibility in Africa for the hepatitis B virus.



Fig. 3. Calibration plot for TREAT-B for indicating treatment eligibility based on the EASL guidelines. EASL, European Association for the Study of the Liver; TREAT-B, treatment eligibility in Africa for the hepatitis B virus.

Compared to the international treatment criteria,^{8–10} the TREAT-B score has many advantages and perfectly fits the resource-limited settings for the following reasons. First, both ALT and HBeAg measurements are widely available in LMICs, and their total cost (<US\$ 10) is much lower than the cost of the conventional tests required to indicate treatment eligibility (Fibroscan, real-time PCR and ALT: >US\$ 60).³ Indeed, both tests have been included in the recently published WHO's list of

essential in vitro diagnostics.³⁶ Second, unlike the treatment algorithms presented in the international guidelines (Table S1), the TREAT-B score is user-friendly and may enable a task shifting from hepatologists to non-specialist doctors or even nurses where there is a shortage of health-care professionals. Third, in contrast to the WHO treatment criteria for LMICs which necessitate several medical visits for ALT measurement, TREAT-B only requires one blood sampling. We are currently evaluating whether those ineligible for antiviral therapy following a clinical evaluation at a single point in time would remain so in the next few years. Fourth, there are validated point-ofcare assays for both HBeAg detection³⁷ and ALT measurement,³⁸ and their use may further facilitate the decentralization of HBV treatment programs. Finally, the TREAT-B is very flexible for its use; the cut-off value can be adapted to the local context, as presented (Table S3). Instead of using the cut-off of ≥ 2 to maximize the sum of sensitivity (82-85%) and specificity (77-78%), applying the score of \geq 3 (sensitivity: 48–53%; specificity: 96%) may be justified where the resources are severely limited; this will restrict the number of patients under treatment by minimizing those unnecessarily treated lifelong.

As a public health intervention to reduce HBV-related mortality in Africa, the role of hepatitis B vaccination alone is limited because an estimated 6.1% of African adults who established chronic infection before the introduction of hepatitis B vaccines are at high risk of dying from HBV-related liver diseases in the next few decades.^{3,6,15} Thus, an additional population-wide "screen-and-treat" may efficiently reduce these deaths, as suggested by a recent modelling study.³⁹ Its feasibility and cost-effectiveness have been demonstrated by the PROLIFICA program in The Gambia.^{6,7} TREAT-B, a simple and inexpensive treatment eligibility score validated in African patients, may facilitate the scale up and decentralization of HBV treatment programs in sub-Saharan Africa, which may ultimately contribute towards the global HBV elimination.

Our study has some limitations. First, it is based on crosssectional data in contrast to the large-scale population-based longitudinal cohort studies in Asia.²¹ Developing the model to predict the hard endpoints such as HCC or liver-related death could provide more direct evidence to assist physicians in selecting those who would benefit most from antiviral treatment, as has been suggested by Tong *et al.*⁴⁰ Obtaining such data, however, requires a long-term observation of African people with chronic HBV infection without giving antiviral therapy, which would not be ethical since the advent of effective antivirals. Moreover, a historical study conducted in The Gambia demonstrated that the persistence of HBeAg, or persistent elevation of viral load (\geq 2,000 IU/ml) or ALT (\geq 40 IU/L) during the 27 years of follow-up were all independently associated with the development of significant fibrosis,¹⁵ supporting the appliTable 4. Performance of TREAT-B, WHO, and REACH-B to select patients eligible for antiviral therapy in derivation (n = 804) and validation set (n = 327).

Derivation set									
		EASL		AASLD			APASL		
	TREAT-B*	WHO	REACH-B [*]	TREAT-B [*]	WHO	REACH-B*	TREAT-B*	WHO	REACH-B
AUROC (95% CI)	0.88 (0.83-0.93)	0.70 (0.65–0.75)	0.90 (0.87-0.94)	0.89 (0.84-0.94)	0.71 (0.66–0.76)	0.90 (0.86-0.94)	0.87 (0.83-0.91)	0.75 (0.71–0.78)	0.84 (0.79–0.89)
p value"	n.a.	< 0.01	0.2	n.a.	< 0.01	0.4	n.a.	<0.01	0.3
Sen (%)	79	86	91	80	88	89	74	94	75
Spe (%)	88	54	80	88	54	79	90	56	80
PLR	6.8	1.9	4.5	6.8	1.9	4.3	7.1	2.1	3.7
NLR	0.2	0.3	0.1	0.2	0.2	0.1	0.3	0.1	0.3
PABAK (95% CI)	0.75 (0.69–0.81)	0.13 (0.08–0.17)	0.61 (0.54–0.68)	0.75 (0.69–0.81)	0.13 (0.08–0.17)	0.60 (0.53-0.67)	0.76 (0.70-0.82)	0.19 (0.14–0.24)	0.59 (0.52–0.66)
Validation set									

	EASL			AASLD			APASL		
	TREAT-B*	WHO	REACH-B	TREAT-B [*]	WHO	REACH-B*	TREAT-B*	WHO	REACH-B [*]
AUROC (95% CI)	0.85	0.65	0.81	0.83	0.67	0.79	0.85	0.67	0.80
	(0.79-0.91)	(0.60 - 0.70)	(0.75-0.87)	(0.77-0.89)	(0.62-0.71)	(0.73-0.85)	(0.80 - 0.90)	(0.63-0.72)	(0.74-0.86)
p value"	n.a.	< 0.01	0.2	n.a.	< 0.01	0.2	n.a.	< 0.01	0.07
Sen (%)	85	90	93	82	92	89	83	94	91
Spe (%)	77	40	38	78	41	38	78	41	38
PLR	3.7	1.5	1.5	3.8	1.6	1.4	3.8	1.6	1.5
NLR	0.2	0.3	0.2	0.2	0.2	2.3	0.2	0.2	0.3
PABAK (95% CI)	0.57	-0.03	-0.05	0.58	0.03	-0.04	0.58	0.03	-0.03
	(0.47-0.68) (-0.06 to -0.01) ((-0.09 to 0.00)	(0.47 - 0.69)	(-0.01 to -0.06)	(-0.08 to 0.00)	(0.48 - 0.69)	(-0.01 to -0.07)	(-0.07 to -0.01)

AASLD, American Association for the Study of Liver Diseases; APASL, Asian Pacific Association for the Study of the Liver; AUROC, area under the receiver operating characteristic curve; NLR, negative likelihood ratio; EASL, European Association for the Study of the Liver; PABAK, prevalence-adjusted and bias-adjusted kappa; PLR, positive likelihood ratio; Sen, sensitivity; Spe, specificity; REACH-B, risk estimation for hepatocellular carcinoma in chronic hepatitis B; TREAT-B, treatment eligibility in Africa for the hepatitis B virus; WHO, World Health Organization.

*The cut-off of 2/4 points for TREAT-B and 6/17 for REACH-B were applied to estimate the sensitivity and specificity.

** p value comparing with AUROC of TREAT-B.



Fig. 4. Receiver operating characteristic curves for TREAT-B, WHO, and REACH-B to indicate EASL treatment eligibility. EASL, European Association for the Study of the Liver; REACH-B, Risk estimation for hepatocellular carcinoma in chronic hepatitis B; TREAT-B, treatment eligibility in Africa for the hepatitis B virus; WHO, World Health Organization. (This figure appears in colour on the web.)

cability of the current international treatment guidelines (that are mostly based on the Asian, European and North American data) to the African context. Secondly, although our study showed good performance of a diagnostic score without requiring HBV DNA to guide treatment initiation in African patients, HBV DNA may be still important in monitoring treatment response. Complete viral suppression, defined by undetectable HBV DNA under treatment, is the key indicator of good treatment response. Although the majority of patients treated with second generation of nucleos(t)ide analogues (entecavir or tenofovir) achieve complete viral suppression,⁶ viral load assay may still be valuable in assessing adherence to treatment.⁴ We will assess the adequacy of monitoring the treatment response without HBV DNA through the PROLIFICA cohort. Thirdly, in this analysis we have excluded patients co-infected with HCV, HDV or HIV. Therefore, the use of TREAT-B in a setting with a high co-infection rate should be further validated. Fourthly, in our cohort only a few individuals were obese or had high alcohol consumption. This limits the generalizability of the score to HBV-infected persons with these additional risk factors, since ALT elevation may be due to a liver process other than that caused by HBV-induced inflammation or fibrosis. Similarly, in this analysis the patients had either HBV genotype A or E. The applicability of the TREAT-B to other places where different genotypes circulate (genotypes B and C in Asia and the Pacific Islands; genotypes D in South and East Africa) needs to be examined. Finally, the number of individuals eligible for treatment was relatively small in the derivation set (n = 58), which might have led to a limited power of the analysis to identify important predictors. Nevertheless, an increase in power through merging the derivation and validation set leads to a similar model composed of ALT and HBeAg as covariates, and this supported the robustness of our model.

In conclusion, TREAT-B represents a promising simple and low-cost diagnostic score that can assist physicians to easily identify HBV-infected individuals in need of treatment in Africa. Its use may contribute towards global HBV elimination by facilitating the scale up and decentralization of HBV treatment programs in LMICs. TREAT-B deserves to be further validated in other African and non-African patients with chronic hepatitis B in LMICs.

Financial support

The PROLIFICA study in The Gambia was supported by the Gambian Ministry of Health and Social Affairs, the MRC Unit The Gambia, the National Public Health Laboratory, the European Commission (European Community's Seventh Framework Program, FP7 grant 265994) and the Gilead Sciences (USA) for providing tenofovir treatment. The study in Senegal was supported by the Agence National de Recherche sur le Sida et les Hépatites Virales (ANRS).

Conflict of interest

MT has received grants from Gilead Sciences. The other authors declare no conflict of interest.

Please refer to the accompanying ICMJE disclosure forms for further details.

Authors' contributions

YS and ML designed the study and drafted the manuscript, and all the authors reviewed and approved it. YS, RN, GN, BS, IB, PS, AC, AJ, HFN, SN, UDA, IC, MM, MT and ML were responsible for the Gambian cohort; MV and PSM for Senegalese cohort; PB and RS for Burkinabé cohort; JN and VL for the cohort in Grenoble; JB for the cohort in Paris; PI and GP for the cohort in Berlin; ML and MT for the cohort in London; and YS for statistical analysis.

Acknowledgements

We thank all the study participants and the PROLIFICA team in The Gambia: in particular Famara Bojang, Mavis Foster-Nyarko, Debbo Jallow, Abdullah Jatta, Sheriff Kolley, Saydiba Tamba, Yamundow Jallow Samba, Alagie Sanneh, Demba Sonko, Lamin Bojang, and Mamina Bojang. We thank the Imperial Biomedical Research Center.

Supplementary data

Supplementary data associated with this article can be found, in the online version, at https://doi.org/10.1016/j.jhep.2018.05. 024.

References

Author names in bold designate shared co-first authorship

- [1] Stanaway JD, Flaxman AD, Naghavi M, Fitzmaurice C, Vos T, Abubakar I, et al. The global burden of viral hepatitis from 1990 to 2013: findings from the Global Burden of Disease Study 2013. Lancet 2016;388:1081–1088. https://doi.org/10.1016/S0140-6736(16)30579-7.
- [2] WHO. Global health sector strategy on viral hepatitis 2016– 2021. Switzerland: Geneva; 2016.
- [3] WHO. Global hepatitis report, 2017. Switzerland: Geneva; 2017.
- [4] WHO. Guidelines on hepatitis B and C testing. Switzerland: Geneva; 2017.
- [5] Njai HF, Shimakawa Y, Sanneh B, Ferguson L, Ndow G, Mendy M, et al. Validation of rapid point-of-care (POC) tests for the detection of hepatitis B surface antigen (HBsAg) in field and laboratory settings in The Gambia, West Africa. J Clin Microbiol 2015;53:1156–1163. <u>https://doi.org/ 10.1128/JCM.02980-14</u>.
- [6] Lemoine M, Shimakawa Y, Njie R, Taal M, Ndow G, Chemin I, et al. Acceptability and feasibility of a screen-and-treat programme for hepatitis B virus infection in The Gambia: the Prevention of Liver

Fibrosis and Cancer in Africa (PROLIFICA) study. Lancet Glob Heal 2016;4:e559-e567. https://doi.org/10.1016/S2214-109X(16)30130-9.

- [7] Nayagam S, Conteh L, Sicuri E, Shimakawa Y, Suso P, Tamba S, et al. Costeffectiveness of community-based screening and treatment for chronic hepatitis B in The Gambia: an economic modelling analysis. Lancet Glob Heal 2016;4:e568–e578. <u>https://doi.org/10.1016/S2214-109X(16)</u> 30101-2.
- [8] European Association for the Study of the Liver. EASL clinical practice guidelines: Management of chronic hepatitis B virus infection. J Hepatol 2012;57:167–185. <u>https://doi.org/10.1016/j.jhep.2012.02.010</u>.
- [9] Terrault NA, Bzowej NH, Chang K-M, Hwang JP, Jonas MM, Murad MH. AASLD guidelines for treatment of chronic hepatitis B. Hepatology 2016;63:261–283. <u>https://doi.org/10.1002/hep.28156</u>.
- [10] Sarin SK, Kumar M, Lau GK, Abbas Z, Chan HLY, Chen CJ, et al. Asian-Pacific clinical practice guidelines on the management of hepatitis B: a 2015 update. Hepatol Int 2016;10:1–98. <u>https://doi.org/10.1007/s12072-015-9675-4</u>.
- [11] Lemoine M, Eholié S, Lacombe K. Reducing the neglected burden of viral hepatitis in Africa: strategies for a global approach. J Hepatol 2015;62:469–476. <u>https://doi.org/10.1016/i.jhep.2014.10.008</u>.
- [12] Andriamandimby SF, Olive M, Shimakawa Y, Rakotomanana F, Razanajatovo IM, Andrianinarivomanana TM, et al. Prevalence of chronic hepatitis B virus infection and infrastructure for its diagnosis in Madagascar: implication for the WHO's elimination strategy. BMC Public Health 2017;17:636. <u>https://doi.org/10.1186/s12889-017-4630-z</u>.
- [13] WHO. Guidelines for the prevention, care and treatment of persons with chronic hepatitis B infection. Switzerland: Geneva; 2015.
- [14] Yang HI, Yuen MF, Chan HL, Han KH, Chen PJ, Kim DY, et al. Risk estimation for hepatocellular carcinoma in chronic hepatitis B (REACH-B): development and validation of a predictive score. Lancet Oncol 2011;12:568–574. <u>https://doi.org/10.1016/S1470-2045(11)70077-8</u>.
- [15] Shimakawa Y, Lemoine M, Njai HF, Bottomley C, Ndow G, Goldin RD, et al. Natural history of chronic HBV infection in West Africa: a longitudinal population-based study from The Gambia. Gut 2016;65:2007–2016. <u>https://doi.org/10.1136/gutinl-2015-309892</u>.
- [16] Lemoine M, Shimakawa Y, Njie R, Njai HF, Nayagam S, Khalil M, et al. Food intake increases liver stiffness measurements and hampers reliable values in patients with chronic hepatitis B and healthy controls: the PROLIFICA experience in The Gambia. Aliment Pharmacol Ther 2014;39:188–196. https://doi.org/10.1111/apt.12561.
- [17] Ghosh S, Sow A, Guillot C, Jeng A, Ndow G, Njie R, et al. Implementation of an in-house quantitative real-time polymerase chain reaction method for Hepatitis B virus quantification in West African countries. J Viral Hepat 2016;23:897–904. <u>https://doi.org/10.1111/jvh.12561</u>.
- [18] Lemoine M, Shimakawa Y, Nayagam S, Khalil M, Suso P, Lloyd J, et al. The gamma-glutamyl transpeptidase to platelet ratio (GPR) predicts significant liver fibrosis and cirrhosis in patients with chronic HBV infection in West Africa. Gut 2016;65:1369–1376. <u>https://doi.org/ 10.1136/gutjnl-2015-309260</u>.
- [19] Boursier J, Zarski JP, de Ledinghen V, Rousselet MC, Sturm N, Lebail B, et al. Determination of reliability criteria for liver stiffness evaluation by transient elastography. Hepatology 2013;57:1182–1191. <u>https://doi.org/ 10.1002/hep.25993</u>.
- [20] Wong VWS, Chan SL, Mo F, Chan TC, Loong HHF, Wong GLH, et al. Clinical scoring system to predict hepatocellular carcinoma in chronic hepatitis B carriers. J Clin Oncol 2010;28:1660–1665.
- [21] Yang H-I, Sherman M, Su J, Chen P-J, Liaw Y-F, Iloeje UH, et al. Nomograms for risk of hepatocellular carcinoma in patients with chronic hepatitis B virus infection. J Clin Oncol 2010;28:2437–2444. <u>https://doi.org/10.1200/IC0.2009.27.4456</u>.
- [22] Wong GL-H, Chan HL-Y, Wong CK-Y, Leung C, Chan CY, Ho PP-L, et al. Liver stiffness-based optimization of hepatocellular carcinoma risk score in patients with chronic hepatitis B. J Hepatol 2014;60:339–345. <u>https:// doi.org/10.1016/j.jhep.2013.09.029</u>.
- [23] Shimakawa Y, Bah E, Wild CP, Hall AJ. Evaluation of data quality at the Gambia National Cancer Registry. Int J Cancer 2013;132:658–665. <u>https://doi.org/10.1002/ijc.27646</u>.
- [24] Mbaye PS, Sarr A, Sire J-M, Evra M-L, Ba A, Daveiga J, et al. Liver stiffness measurement and biochemical markers in Senegalese chronic hepatitis B patients with normal ALT and high viral load. PLoS One 2011;6:e22291. <u>https://doi.org/10.1371/journal.pone.0022291</u>.
- [25] Bonnard P, Sombié R, Lescure F-X, Bougouma A, Guiard-Schmid JB, Poynard T, et al. Comparison of elastography, serum marker scores, and histology for the assessment of liver fibrosis in hepatitis B virus (HBV)infected patients in Burkina Faso. Am J Trop Med Hyg 2010;82:454–458. <u>https://doi.org/10.4269/ajtmh.2010.09-0088</u>.

Research Article

- [26] Sullivan LM, Massaro JM, D'Agostino RB. Presentation of multivariate data for clinical use: The Framingham Study risk score functions. Stat Med 2004;23:1631–1660. <u>https://doi.org/10.1002/sim.1742</u>.
- [27] Altman DG, Vergouwe Y, Royston P, Moons KGM. Prognosis and prognostic research: validating a prognostic model. BMJ 2009;338:b605.
- [28] Byrt T, Bishop J, Carlin JB. Bias, prevalence and kappa. J Clin Epidemiol 1993;46:423–429.
- [29] Bossuyt PM, Reitsma JB, Bruns DE, Gatsonis CA, Paul P, Irwig L, et al. STARD 2015: an updated list of essential items for reporting diagnostic accuracy studies. BMJ 2015;351:h5527. <u>https://doi.org/10.1136/bmj. h5527</u>.
- [30] Jaquet A, Nouaman M, Tine J, Tanon A, Anoma C, Inwoley A, et al. Hepatitis B treatment eligibility in West Africa: uncertainties and need for prospective cohort studies. Liver Int 2017;37:1116–1121. <u>https://doi. org/10.1111/liv.13484</u>.
- [31] Aberra H, Desalegn H, Berhe N, Medhin G, Stene-Johansen K, Gundersen SG, et al. Early experiences from one of the first treatment programs for chronic hepatitis B in sub-Saharan Africa. BMC Infect Dis 2017;17:438. <u>https://doi.org/10.1186/s12879-017-2549-8</u>.
- [32] Chang M-H. Natural history and clinical management of chronic hepatitis B virus infection in children. Hepatol Int 2008;2:S28–S36.
- [33] Evans A, Connell APO, Pugh JC, Mason S. Geographic variation in viral load among hepatitis B carriers with differing risks of hepatocellular carcinoma. Cancer Epidemiol Biomarkers Prev 1998;7:559–565.
- [34] Chen CJ, Yang HI, Su J, Jen CL, You SL, Lu SN, et al. Risk of hepatocellular carcinoma across a biological gradient of serum hepatitis B virus DNA level. JAMA 2006;295:65–73. <u>https://doi.org/10.1001/jama.295.1.65</u>.

- [35] Mendy ME, Welzel T, Lesi OA, Hainaut P, Hall AJ, Kuniholm MH, et al. Hepatitis B viral load and risk for liver cirrhosis and hepatocellular carcinoma in The Gambia, West Africa. J Viral Hepat 2010;17:115–122. https://doi.org/10.1111/j.1365-2893.2009.01168.x.
- [36] WHO. World Health Organization model list of essential in vitro diagnostics. First edition (2018). Switzerland: Geneva; 2018.
- [37] Clement F, Dewint P, Leroux-Roels G. Evaluation of a new rapid test for the combined detection of hepatitis B virus surface antigen and hepatitis B virus e antigen. J Clin Microbiol 2002;40:4603–4606.
- [38] Jain S, Rajasingham R, Noubary F, Coonahan E, Schoeplein R, Baden R, et al. Performance of an optimized paper-based test for rapid visual measurement of alanine aminotransferase (ALT) in fingerstick and venipuncture samples. PLoS One 2015;10:1–15. <u>https://doi.org/ 10.1371/journal.pone.0128118</u>.
- [39] Nayagam S, Thursz M, Sicuri E, Conteh L, Wiktor S, Low-Beer D, et al. Requirements for global elimination of hepatitis B: A modelling study. Lancet Infect Dis 2016;16:1399–1408. <u>https://doi.org/10.1016/S1473-3099(16)30204-3</u>.
- [40] Tong MJ, Hsien C, Hsu L, Sun H-E, Blatt LM. Treatment recommendations for chronic hepatitis B: an evaluation of current guidelines based on a natural history study in the United States. Hepatology 2008;48:1070–1078. <u>https://doi.org/10.1002/hep.22476</u>.